Author's response to reviews

Title: Health Related Quality of Life among Iraqi Immigrants Settled in Malaysia

Authors:

Aqil M Daher (aqil702001@yahoo.com)
Hisham S Ibrahim (hlbe2004@yahoo.com)
Thaaer M Daher (thaeer_alsad@yahoo.com)
Ali K Anbori (dr.alianbori@gmail.com)

Version: 5 Date: 22 May 2011

Author's response to reviews: see over
22.05.2011

The Editor,
BMC Public Health Journal

Dear Sir/Madam,


With reference to the above manuscript and on behalf of the co-authors, I am submitting the revised version of the manuscript addressing the comments of the reviewer. I would like to extend our sincere appreciation to all reviewers whose comments have contributed to substantial improvements in the manuscript. Our responses to the comments (language editing) are included in the manuscript

The authors once again wish to thank the reviewers for their valuable criticism, which has been educational for us. We do hope the revisions made so far meet your requirements and we look forward to your considered response.

Thank you

Yours truly

Dr Aqil Mohammad Daher